BANCA D'ITALIA

Temi di discussione

del Servizio Studi

The interaction between face-to-face and electronic delivery: the case of the Italian banking industry

by E. Bonaccorsi di Patti, G. Gobbi and P. E. Mistrulli



Number 508 - July 2004

The purpose of the Temi di discussione series is to promote the circulation of working papers prepared within the Bank of Italy or presented in Bank seminars by outside economists with the aim of stimulating comments and suggestions.

The views expressed in the articles are those of the authors and do not involve the responsibility of the Bank.

Editorial Board:

Stefano Siviero, Emilia Bonaccorsi di Patti, Fabio Busetti, Andrea Lamorgese, Monica Paiella, Francesco Paternò, Marcello Pericoli, Alfonso Rosolia, Stefania Zotteri, Raffaela Bisceglia *(Editorial Assistant)*.

THE INTERACTION BETWEEN FACE-TO-FACE AND ELECTRONIC DELIVERY: THE CASE OF THE ITALIAN BANKING INDUSTRY

by Emilia Bonaccorsi di Patti, Giorgio Gobbi and Paolo Emilio Mistrulli*

Abstract

We empirically investigate the relevance of demand-side complementarity between electronic and traditional provision of banking services. Since no systematic data on prices for the two types of services is available, it is not possible to estimate cross-elasticities of demand. We resort to two indirect tests. The first test is based on estimating the relationship between branches and the diffusion of e-banking services in local markets, controlling for individual bank and market characteristics employing new data for Italian banks referring to 1998-2001. We find that banks expanded relatively more in the e-business in those local markets where they had relatively fewer branches, with the exclusion of markets where the banks were chartered. The second test is based on measuring the impact of the joint provision of banking services - electronically and at traditional branches - on banks' revenues per customer. We estimate a non-standard revenue function that relates revenues from asset management, brokerage and payment services to the share of customers employing e-banking, given the total number of bank customers. Our results show that a high share of e-customers is associated with a reduction in revenues per customer. This evidence suggests that banks did not extract substantial consumer surplus from the joint provision of electronic services and traditional services at the branch. We interpret the results of both our test as not consistent with the hypothesis of complementarity between stores and e-commerce in the banking industry.

JEL classification: D12, G21, O32.

Keywords: commercial banks, e-banking, electronic transactions, store proximity.

1. Introduction	7
2. E-Banking in Italy: Stylized Facts	10
3. Theoretical and Methodological Issues	
4. An Empirical Model of the Diffusion of E-Banking	
4.1 Results	
5. The Impact of Electronic Provision on Revenues	
5.1 The Empirical Model: A Non-Standard Revenue Function	
5.2 Results	30
6. Conclusion	
Tables	
References	45

* Banca d'Italia, Economic Research Department.

1. Introduction¹

Recent developments in information processing, telecommunications and related technologies (IT) have significantly changed the way firms sell their products and services. Retail banking is an example of an industry that has been deeply affected by the evolution of IT. Since the early 1980s an increasing volume of cash management services has been serviced through automated teller machines (ATMs) and cash dispensers. Several countries have experienced a widespread adoption of point-of-sale terminals for electronic funds transfer (POSes) and other electronic cash-less instruments for retail payments (BIS, 2003). Home-banking services through private proprietary networks were launched, although with mixed success (Radecki, Wenninger and Orlow, 1997).

In the late 1990s one of the most important innovations in the financial industry was the provision of information and transaction financial services over the Internet. The change from restricted proprietary systems to a widely adopted open network has greatly enhanced the potential development of remote banking activities.² The standardization provided by the world-wide-web has cut the communication costs of reaching customers electronically, stimulating the adoption of new marketing and commercial strategies.

The consensus view is that in the medium term the Internet will produce extensive changes in the structure of the banking industry but there are different views on the pattern of these changes. On the one hand, electronic delivery is expected to increase the degree of competition among financial intermediaries because it lowers entry costs (Clemons and Hitt, 2000). In addition, the standard advertising theory (e.g. Stigler, 1961) suggests that the substantial increase in publicly available information about prices over the Internet should reduce both prices and price dispersion. On the other hand, the Internet offers banks the possibility of developing new

¹ The opinions expressed do not necessarily reflect those of the Banca d'Italia or its staff. We thank Mattew Shum and two anonymous referees for their comments and suggestions.

² Remote banking is usually defined as the provision of banking services without face-to-face contact between the bank employee and the customer (ECB, 1999). Generally speaking, remote banking services include self-banking such as multi-purpose ATMs, telephone banking and PC or electronic banking. In this paper we use the term remote customer to indicate customers connected either by phone or computer.

products, based on the use of personal information on individual customers at relatively low costs (Claessens, Glaessner and Klingebiel, 2000). A consequence of product tailoring and price discrimination could be a reduction in competition.

Making more accurate predictions is difficult because of the lack of data on the provision of e-banking services. In particular, it is unclear to what extent e-banking services are substitutes or complements with respect to traditional transactions delivered at branches from the point of view of customers. A second issue is the impact of different pricing strategies on consumer choices and, ultimately, on bank profits, especially when there are cost rigidities due to difficulties in cutting the branch network. Banks have combined traditional and electronic operations to different degrees. At one end of the spectrum, some banks provide only remote services, either on the phone or the Internet; at the other, many institutions have chosen not to provide remote services at all. The most common strategy is to offer a limited set of information or transaction services electronically together with traditional branch transactions (the so called "click and mortar" strategy).

Such variety of choices corresponds to the complex pattern of supply and demand forces, which has driven the first wave of Internet banking. On the supply side, e-banking is characterized by a large initial investment and substantial reputation advantages. These factors imply scale economies and very low marginal costs. Pure "virtual banks" have relied on aggressive pricing to build up a large customer base to exploit the economies of scale Econometric evidence shows that pure electronic banks face a longer and flatter learning curve than traditional de novo banks, suggesting that "virtual" banking may not be an inferior business model in the long run (DeYoung, 2001) even if in medium term "virtual" banks are at a disadvantage with respect to their "click and mortar" competitors.

Traditional banks started to offer remote services under the threat of competition by pure electronic banks.³ Given their large customer base, they have been able to afford the large initial investment and cross-subsidize their e-banking operations as part of a global expansion strategy.

³ Surveyed banks supplying e-banking services or intending to do so in the US generally referred to a need to remain competitive and retain customers, rather than increase revenues to cover the costs (Furst, Lang and Nolle 2001, 2002, Sullivan, 2000 and 2001).

The success of the "click and mortar" business model will depend on the importance of economic complementarity between electronic and physical delivery. One opinion largely held in the industry is that demand-side complementarities are substantial because customers perceive electronic access as an additional feature of traditional banking services, thus preferring the "click and mortar" bank (JP Morgan, 2000; Bank for International Settlements, 2001). If complementarities are substantial, banks can extract the surplus generated by the preference of consumers for a single provider with multiple delivery channels and cover the costs of maintaining a branch network. If customers attribute great value to the option of accessing branch services, the surplus that can be extracted from e-banking is larger because competition in e-banking is less effective due to a form of "lock-in".

To date the direct impact of the "click and mortar" strategy on bank profitability has not been encouraging. In Europe traditional banks have developed e-banking services without substantially reducing their branch networks. The result has been that they experienced an increasing cost pressure, which in turn led to a reduction in investments required to further develop their electronic sales (ECB, 2002).

The viability of the "click and mortar" model, in which the provision of services electronically is compatible with large branch networks, depends on consumers' willingness to pay for the availability of a variety of delivery channels from the same bank, i.e. on the extent to which they perceive e-banking and face-to-face services as close complements.⁴ In this study we test if electronic and face-to-face provision of banking services are perceived as complements by consumers in two ways. The first way is to estimate the empirical relationship between the number of e-customers a bank has and the size of its branch network. We employ a large sample of "click and mortar" banks and focus on their e-banking connections in a number of local markets. The second way is to assess the impact of the provision of e-banking services on revenues of banks that have also a branch network. We estimate a non-standard revenue function relating bank revenues not only to output and input quantities but also to the number of customers with e-banking accounts, given the total number of bank customers.

Complementarity between the electronic provision of goods and traditional stores is largely unexplored since systematic and detailed data on prices and use of electronic services are not available. Such data do not exist for the banking industry either but Italian banks have filed unique information on the number of customers connected electronically, since the late 1990s. The data include information on each bank, the number of customers by type of connection (electronic, phone), type of customer and geographical location. In the analysis presented below, we employ this data for the first time. The first test is performed on bank-market observations, hence we control for differences in local demand by including local market variables and focus on the effects of bank i in market j branch density on the number of e-customers. In the second test the bank-level information on the number of customers is matched with the balance sheets and income statements of banks and other banking data from Supervisory Reports to estimate the revenue function.

Results from the first test show that banks expanded relatively more in the e-business in those local markets where they had relatively less branches, with the exclusion of markets where the banks were chartered. Consistently, our second test suggest that banks with a high share of e-customers had lower revenues per customer. We interpret these results as not supportive of the hypothesis that face-to-face services and e-banking are complements.

The paper is structured as follows. Section 2 illustrates the data and describes the main features of electronic banking in Italy. Section 3 discusses the main issues and summarizes the previous literature. Section 4 presents the empirical model of the relationship between e-banking and bank branches; Section 5 contains the analysis of revenues. Section 6 concludes.

2. E-Banking in Italy: Stylized Facts

Data on e-banking and other remote connections (e.g. phone) have been collected systematically by the Bank of Italy since 1996. The data include the number of customers, sector

⁴ The notion of complementarity we refer to is more general than the text book definition and is close to that of scope economies in consumption (e.g. Berger, Humphrey and Pulley, 1996).

of economic activity (household or corporate) and location⁵. To our knowledge, the information is the most comprehensive at the international level because it combines geographical with banklevel detail. Customers are classified according to their use of the remote connections between those who only have access to information and those who have actually performed at least one transaction. Banks also report the number of customers that use phone banking or other services, for example those that a bank offers to its employees through the internal network.

Our analysis focuses only on households because the aim of the paper is to study the interaction between physical and electronic delivery in the retail market, that is the market for banking services to small businesses and households. As our data do not allow us to distinguish small businesses from large firms, we focus on households.

Descriptive statistics on Italian e-banking are reported in Tables 1-2. The data reported refer to the years 1998-2001. As shown in Table 1, the adoption of phone and e-banking – also known as home banking - by banks has increased substantially in the last three years among all bank size categories.

At the end of 2001, 529 banks supplied e-banking services to households. The number of banks offering phone banking services in 2001 was much lower (186) than of those offering e-banking services. The expansion of e-banking was relatively slow until 1999. In 2000 and 2001 the rate of growth was higher than in the previous years, especially for e-banking services. This was partly due to the entry of a number of small banks belonging mostly to banking groups and highly focused on e-banking services supplied through the Internet. As reported in Table 2, at the end of 1998 only 0.2 per cent of households could access their bank via their PC and perform some transactions. In 2001 this percentage was 7.6. The share of phone banking accounts was significantly higher at the beginning of the period (3.5 per cent) but increased less rapidly afterwards. In 2001, 7.2 per cent of households could access their bank by phone calls for transactional purposes.

⁵ Customers are classified by province of residence. In Italy there are 103 provinces. According to the antitrust regulation deposit markets for households are considered as large as the province.

Additional information on the electronic supply of financial services in Italy is contained in the report on the annual survey of Internet banking conducted by the Bank of Italy, starting in 2000. The survey covers banks and non-bank financial intermediaries. The number of non-bank financial intermediaries that offer electronic services is small (Banca d'Italia, 2001a). Within diversified financial conglomerates, electronic services are mostly provided by banks because they have the technology for collecting and processing large amounts of data and because the financial services provided electronically are usually bundled with a checking account. In the empirical analysis presented below we only include banks. Nevertheless, we do not expect a large bias due to the exclusion of financial companies.

According to the survey, e-banking has affected only the deposit side of the market and the brokerage/trading business. Banks typically supply a package including the typical set of information and transaction services tied to a checking account (money transfers and other payment services) plus brokerage services. Although information services on loans are provided electronically on the banks' web sites or on portals that compare products of different institutions, loans are not supplied electronically.

	Investment and asset management	Payment services	Other services	Total
Millions of euros	79.5	7.5	9.2	96.2
As a percentage of the Corresponding Industry revenues	0.8	0.3	-	0.7

Box A: Revenues from Internet transactions of Italian banks in the year 2000

Source: Bank of Italy Supervision Bulletin (Banca d'Italia, 2001a) for revenues from Internet transactions and Annual Report on 2000 for total industry revenues (Banca d'Italia, 2001b). Total industry figures are net revenues, i.e. they are obtained as the difference between commissions and fees receivable and commissions and fees payable. Total revenues from Internet transactions have been taken as a percentage of industry net income from services.

Investment and asset management services account for almost 80 per cent of revenues generated by e-banking, while the provision of payment services contributes to less than 8 per cent. Total revenues from Internet transactions amount to less than 1 per cent of total bank net

revenues from commissions and fees. Importantly, the contribution of e-banking to income from services varies considerably among the banks responding to the survey. About one in ten of the banks in the sample reported revenues amounting to more than 2 per cent of total revenues from services and another fifth fell in the range between 1 and 2 per cent. Although no comparable data are available for year 2001, it is reasonable to expect that both the level of revenues from electronic transactions and their dispersion among banks should have increased owing to the increase in the number of customers and active banks. Thus, the data should contain enough information to perform the statistical tests based on the revenue function described in section 5.

3. Theoretical and Methodological Issues

The introduction of e-banking modifies the way banks can enter new markets and reach customers. To our knowledge, no specific theoretical model studies the interaction between traditional (i.e. face-to-face) provision and electronic provision of services in the banking industry, although a number of papers have emphasized the role of accessibility of services to customers in shaping banks' strategies. Accessibility is usually defined as the cost for customers to reach a bank branch as a combination of distance, which entails transportation costs, time to complete a transaction and monetary expenses (Evanoff, 1988). Different combinations of accessibility attributes determine different degrees of convenience, which is valued by customers when they choose where or how to purchase a given financial service. Convenience has been a primary strategic variable for banks competing in markets where regulation prevents price flexibility, as in the case of regulation Q in the United States. More recently, Dick (2003) finds empirical evidence that the staffing and geographic density of local branches remain important determinants in consumers' choices of a depositary institution, even when there are no constraints on price setting.

In general terms, increasing convenience is a way of raising consumers' surplus. If one considers a single highly standardized service, such as the purchase of securities or a money transfer, an electronic transaction is a substitute for a trip to the branch. The two services imply different combinations of accessibility attributes (time, distance, connection costs), ease of use and price. The costs of accessing banking services electronically include investing in the

specialized knowledge needed to use a computer and the necessary software. Knowledge barriers may limit the size of the market to a subset of bank customers in the short term, but tend to weaken at the pace of IT diffusion among consumers. The consensus view is that the convenience of accessing financial services electronically largely outweighs that of accessing it at the branch for computer-literate consumers (Mishkin and Strahan, 1999).

On the bank side, the marginal costs of delivering highly standardized services electronically are far lower than those of delivering them physically. Although systematic evidence is scarce, data from case studies suggests that the difference is at least of one order of magnitude (Sato and Hawkins, 2001). Consequently, competition should drive down prices, letting consumers appropriate of the surplus (convenience) generated by banking on-line. Two factors may prevent competition allowing banks to extract the surplus. The first is that even highly standardized financial services are not standardized enough to be perceived as equivalent to a service supplied at a branch. For example, a significant proportion of the information exchanged between bank employees/officers and customers may be non-standardized (soft information) and therefore most of on-line transactions are part of a bundle of services that includes some kind of physical interaction.⁶

The second factor in determining the magnitude of the surplus that the bank can seize is the relative importance of cross-selling. The bundle of services provided electronically is usually not the same as the one available at a branch. For this reason it is difficult to assess a priori whether electronic delivery competes with branches, even within the same banking organization, or whether it is a way to increase output volumes and banks' profits by selling different baskets of products. For example, in the case of Italy the vast majority of services provided electronically are either brokerage services or payment services, whereas the full set of services offered at a branch is much larger. In the US banks also supply some types of standardized loans

⁶ This view seems to be rather common within the banking industry. For example, in Italy some banks have complemented their e-banking facilities with a network of financial salesmen. The distinction between soft and hard information is not independent from the state of the art communication technologies. A fairly general discussion of the issue can be found in Leamer and Storper (2001).

electronically (Berger 2003); in the Scandinavian countries virtually all banking services can be delivered through the Internet (European Central Bank, 2002).

Banks can successfully implement cross-selling strategies if consumers are willing to pay for one-stop banking (scope economies in consumption), namely they experience synergies in consuming different financial services from the same bank. Consumers' synergies in banking have been extensively discussed in Berger, Humphrey and Pulley (1996). The argument is that economies of scope in consumption may arise from reductions in transaction, transportation and search costs. The exchange of soft information may also play a role. For example, personal knowledge and reputation with a bank generated by a deposit account may increase the probability of favorable conditions on a future loan application. Customers would prefer to hold an account with a bank that jointly supplies payments, deposit services and loans, which is currently not the case for pure e-banks in many countries, including Italy. Even if banks do not supply loans over the Internet it is likely that soft information remains important (Berlin and Mester, 1999; Kashyap, Rajan and Stein, 2001).

A full evaluation of the relevance of scope economies in consumption in shaping the developments of e-banking would require detailed information on prices charged by different banks for similar services delivered through different channels, as well a comparison between pricing strategies both of pure Internet banking and "click and mortar" banks.

Ideally, one would want to study the effect of prices for both services on the banks' market shares to measure the cross-elasticities of the demand faced by each bank in the two segments. Unfortunately, data on prices are not available on a systematic basis. In the analysis discussed below we follow two indirect avenues. First, we estimate an empirical relationship between ebanking diffusion and branches by banks in different local markets. We exploit the fact that the supply of e-banking facilities has no geographical boundaries and that banks cannot differentiate their policies and prices across customers located in different areas. Marketing strategies for online services and prices cannot be diversified across geographic markets. On the contrary, face-toface sale through branches has a rather limited geographical reach. We argue, therefore, that once a bank has taken the decision of entering the e-banking business, the location of its customers will be by and large out of its control and will depend on market-specific demand characteristics. To the extent that branches matter to e-customers, also the density of the bank's branches in each local market should play an important role in how market demand is allocated across banks.

If economies of scope in consumption are important we should observe that "click and mortar" banks are more likely to expand their e-banking business in the same geographical markets where they have a high branch density, servicing the customers they already have or increasing their market share in both segments by exploiting customers' preference for joint accessibility. On the contrary, if complementarities in consumption are relatively unimportant, banks will attract e-banking customers mainly where they do not have branches because of the lower prices. A third possibility is that the set of customers consuming financial services on-line is disjoint from the set of those using physical channels, so that the two markets are completely separated.

The second indirect test of the effects of scope economies in the consumption of financial services is based on the revenue function model used by Berger, Humphrey and Pulley (1996). In particular we test the effect on bank revenues of an increase in the number of customers using e-banking facilities, holding constant the total number of customers. The intuition is that if there are economies of scope in consumption consumers are willing to pay for the convenience of making transactions electronically with the same bank from which they can purchase face-to-face services at branches. In this case banks are able to extract the consumer surplus generated by the provision of services electronically and their revenues should be positively correlated with the share of consumers that have access to e-banking facilities.

4. An Empirical Model of the Diffusion of E-Banking

In this section we analyze how consumers are distributed across banks in local markets, when intermediaries are active both in the traditional and the e-banking business. In particular, we attempt to verify indirectly whether consumers value the option of physical accessibility to the bank when they have the alternative of connecting to their bank remotely. As discussed below, we focus on banks that are joint producers and exclude pure remote banks. We formulate three hypotheses on the reduced-form relationship between bank branches and the diffusion of e-banking in local geographical markets. These hypotheses result from the interaction of demand

and supply effects, although we expect demand factors prevail in our model. The reason is that the bank fixed effects included in the estimation capture e-banking prices, which are the same across all markets, and any other factor which is fixed at the bank level, for example set-up costs for e-business and reputation. The variation in each bank's capacity to attract e-customers – given market level demand – is a function of the services sold by each bank in that specific market. We focus on branch accessibility as the relevant attribute.

The <u>complements hypothesis</u> states that customers value the joint provision of e-banking and face-to-face banking services from the same provider. Given the general propensity to demand e-banking in a local market, a bank will expand in the e-banking segment relatively more in the same geographical markets where it has many branches. An extensive branch network provides additional value to customers in the e-banking segment because the physical and the electronic delivery channels are perceived as complements. The prediction of the <u>complements</u> <u>hypothesis</u> is that the number of customers connected in a local market is positively related to the size of the branch network in that market, everything else being equal.

The <u>substitution hypothesis</u> posits that consumers using e-banking facilities consider the services provided electronically not very different from those provided at the branches, with the exception of the costs of accessing the service (mainly "shoe-leather" costs for the branches). The cost of connecting electronically to the bank is independent of distance; instead, the cost of reaching a branch is higher the lower the density of branches of the bank in the market. Therefore, the relative convenience of establishing an electronic connection to each bank is greater the lower the density of branches of that bank. This hypothesis predicts that the size of the branch network is negatively related to the expansion of electronic connection for each bank, everything else being equal.

Finally, the <u>segmentation hypothesis</u> states that e-banking and traditional services are two different products and customers consuming one or the other do not attribute any value to the joint provision of the two by the same bank. In this case we should not observe any systematic relationship between branches and the diffusion of e-banking in different markets.

With reference to an empirical model where a measure of e-banking customers is regressed on bank branches, the predictions of each hypotheses are the following:

- <u>i)</u> <u>Complements hypothesis:</u> ∂ E-CUSTOMERS/ ∂ BRANCHES > 0;
- <u>ii)</u> Substitution hypothesis that ∂ E-CUSTOMERS/ ∂ BRANCHES < 0;
- <u>iii</u>) Segmentation hypothesis that ∂E -CUSTOMERS/ $\partial BRANCHES = 0$.

However, if we regress the number of e-customers on the number of branches we will not be able to identify the relationship between branches and e-customers because the coefficient will incorporate the effect that an increase in the number of branches will have on the total number of bank customers. The relationship between face-to-face accessibility (branches) and the demand for electronic access has to be purged by the indirect effect that branches have on the diffusion of electronic accounts via their effect on the number of customers that a bank has in a given local market. The identification of the relationship between branch expansion and e-banking expansion requires instrumental variables estimation.

We construct a simultaneous equations model that specifies the relationship between the number of bank customers with electronic connections (e-customers), the total number of bank customers, bank branches, and control variables for demand and supply conditions at the market level. While it is obvious that the total number of customers influences the number of e-customers, we need to control for a potential reverse causation from the number of e-customers to the total number of customers. The reason is that if an existing customer activates an e-banking account, he/she is counted as one new e-customer in our data but the total number of bank customers does not change. Instead, if a customer of bank *i* opens an electronic account at bank *k*, he or she will be recorded both as a new customer of bank *k* and as a new customer with electronic access of bank *k*. Hence, a bank that expands only in e-banking services expands its total customer base, whereas one that switches its customers to e-banking does not.⁷

We model these relationships with the two-equation simultaneous equations model below, exploiting the availability of data at the bank-market level, for a three-year period. The advantage of bank-market level data is twofold: i) we can control for differences in the demand faced by

⁷ This is why we do not model electronic accounts as a share of total accounts. We want to emphasize the dynamics of both components and not that of their share.

banks in different markets; ii) we can control for most differences across banks with bank fixed effects.

As shown below, equation (1) specifies the number of e-customers of bank *i* in market *j* in period *t* as a function of the total number of customers of bank *i* in market *j* in period *t*, the branch network of bank *i* in market *j* in period *t* and a vector Z_1 of demand and supply market level controls. Equation (2) specifies the total number of customers of bank *i* in market *j* in period *t* as a function of e-customers of bank *i* in market *j* at time *t*, the bank's branches in market *j*, and a vector Z_2 of demand and supply conditions in market *j* in period *t*:

$$E-CUSTOMERS_{IJT} = \alpha_{11} + \beta_{11} \bullet TOTAL \ CUSTOMERS_{IJT} + \beta_{12} \bullet G(BRANCHES)_{IJT}$$
$$+ \Gamma_1 \bullet \mathbf{Z}_1 + \Phi_1 \bullet BANK \ FIXED \ EFFECTS_I + \Psi_1 TIME \ EFFECTS_T + \varepsilon_{11JT}$$
(1)

$$TOTAL \ CUSTOMERS_{IJT} = \alpha_{21} + \beta_{21} \bullet E - CUSTOMERS_{IJT} + \beta_{22} \bullet H(BRANCHES)_{IJT} + \Gamma_2 \bullet \mathbb{Z}_2 + \Phi_2 \bullet BANK \ FIXED \ EFFECTS_I + \Psi_2 TIME \ EFFECTS_T + \varepsilon_{2IJT}$$
(2)

The <u>complements hypothesis</u> predicts a positive coefficient β_{12} , the <u>segmentation</u> <u>hypothesis</u> a nil effect β_{12} , and the <u>substitution hypothesis</u> a negative coefficient β_{12} . Given that the functions *g* and *h* are not know a priori, we test different functional forms (linear and quadratic). We do not exclude a priori the coexistence of more than one effect, with a dominance of one over the other in some range of the size of the branch network. For example, it could be the case that an increase for bank that has very few branches has a positive effect on attracting e-customers, but when the bank has many there is no effect. For this reason we estimate a linear (not reported) and a quadratic function. In both equations the numbers of e-customers, of total customers and branches are transformed in natural logs to reduce heteroscedasticity. Thus, the coefficient β_{12} can be interpreted as an elasticity and its sign identifies which hypothesis holds.

The first endogenous variable, the number of e-customers, is measured by the log of the number of households resident in market j having an Internet account to perform transactions with bank i at the end of year t (LNECUSTMKT). Markets are defined as the 103 Italian provinces, the standard definition of local markets in antitrust policy and empirical research on

Italian data. The second endogenous variable is the log of the end-of-year number of checking accounts (LNTOTCMKT) of each bank in each province. Although a customer could have more than one checking account, we argue that at the local market and bank level this variable should be very close to the number of customers. The key explanatory variable in both equations is the end-of-year log of the number of branches of each bank in each province (LNBRANCHMKT); we include also a second-order term (LNBRANCHMKT_SQ). We denote these three variables with the suffix MKT to distinguish them from their counterparts at the bank level (BK), employed in the revenue analysis in Section 5.

In equation (1) the other variables (the vector Z_1) control for the demand for e-banking services in market *j* and supply conditions. We include also bank fixed effects to capture pricing of e-banking and other factors that do not vary across local markets for the same bank, and time effects to control for the general increase in the diffusion of e-banking over the period studied. The fixed effects are particularly important in controlling the characteristics of e-banking services because they do not vary across markets.

The demand variables are the following. The first is the log of per capita value added (LNVADDED) in the local market, capturing economic development and, most likely, also social development. Since we expect the propensity to use e-banking to be related to human capital we include a variable of human capital (EDUCATION), given by the share of population that holds a high school diploma or higher. The number of Points of Sale (stores connected for direct payment with debit cards) divided by population (1,000) in market *j* at time *t*-*1* is a measure of customers' propensity to use new technologies (POSPOP_1). The variable enters the regression with one lag to avoid endogeneity with respect to the diffusion of e-banking and other new payment technologies. The log of the volume of shares held in households' portfolios at banks in market *j* at time *t*-*1* (LNSHAREMKT_1) controls for differences in the propensity of consumers to buy shares and in the demand for on-line trading services. Since no variable is available at the local market level to capture the diffusion of computers or Internet usage, we resort to a regional-level measure of the share of households that have a personal computer (PCHOUSE). Differences in transportation costs to reach bank branches, given the number of bank branches in the province, are accounted for by the share of population living in areas that are classified as mountains (more

than 600 metres altitude; POPMOUNTAIN). This variable should be positively correlated with transportation costs and have a positive effect on the demand for e-accounts because customers have a greater incentive to avoid going to the branch. A second variable measuring transportation costs is the ratio of kilometers of roads to the area in square kilometres (ROADS). Since no information is available at the province level we resort to regional measures. This variable should proxy for connectivity in the region, once we control for the composition of the region in terms of mountains by POPMOUNTAIN.⁸ We include also the share of population living in the main city (SHURBAN), which should capture network effects in knowledge diffusion and hence have a positive effect on the number of e-customers. The last three variables are observed for one year only.

On the supply-side, competitive conditions in the deposit market are measured by the endof-previous-year Herfindahl index of deposit market concentration, computed at the province level (HERFDEP_1). We do not have a prior on the sign of the coefficient. Customers may demand more e-banking services where there is less competition in face-to-face services, or banks operating in less competitive environments may have the incentive to compete less actively also in the e-banking or not encourage it because it could erode their rents in the face-to-face market.

In equation (2) some of the control variables in Z_2 differ from those in equation (1) because they are supposed to capture the determinants of the demand of bank checking accounts, and are not specific to e-banking. In particular, we include the volume of mortgages to households (LNMORTGAGE) because a household that has a mortgage is very likely to have a checking account to pay the instalments. We argue that this factor should not affect the demand for ebanking because there is no electronic provision of mortgages in Italy. Even if mortgages could be correlated to e-banking because of common factors – education, wealth - we argue that these factors should be controlled for by the other variables. As in equation (1) we include the log of per capita value added (LNVADDED) as a proxy of wealth and economic activity. We include the log of population in the province (LNPOP) that serves as a scaling variable controlling for differences in the size of markets. An urbanization variable should capture the general degree of banking development in the province (SHURBAN); the control for human capital (EDUCATION) is included because the demand of banking services is likely to be related to education. This specification is consistent with the existing evidence on variables that affect the demand for checking accounts (e.g. Caskey, 1994). On the supply side we include the lagged Herfindahl index of deposits also in equation (2) to control for competitive conditions in the local market. Also in equation (2) we control for bank level characteristics with fixed effects. Year dummy variables are included.

The model is identified by exclusion restrictions because the variables included in Z_1 are not the same as those in Z_2 . In particular, Z_1 includes the variables PCHOUSE, LNSHAREMKT_1, POSPOP_1, POPMOUNTAIN and ROADS measuring computer diffusion, volume of securities, propensity to use technology in payments, transportation costs and factors affecting the demand for electronic services but not the demand for checking accounts. These variables should affect the number of bank customers in a given market only to the extent that they influence the demand for electronic accounts and are excluded from equation (2). The vector Z_2 , instead, includes determinants of the number of checking accounts (LNMORTGAGE, LNPOP) not included in equation (1). These variables affect the number of electronic accounts only indirectly, i.e. through their effect on the number of total checking accounts. In the 2SLS estimation they are included in the first stage for both endogenous variables but are excluded from equation (1) in the second stage.⁹

A problem that could arise in our specification is that, even if the supply of e-banking services is fixed across local markets, marketing strategies may be somewhat flexible at the local market level and the bank fixed effects would not control for this factor. In particular, the branch network may be used to advertise the existence and the characteristics of the bank's on-line facilities. In addition, bank reputation may vary across local markets and the "click and mortar"

⁸ Regions with large areas of mountains may have more kilometres of roads simply because of the impossibility of building direct routes between places.

⁹ We estimated the model including also LNSHAREMKT_1 in equation (2) and the results were robust.

banks may promote e-banking by exploiting the strength of their brand name more effectively in local markets where they have a relatively large number of branches. In order to control for differences in reputation we include in both equations the variable ORIGIN, equal to 1 in the local market where each bank was originally chartered, 0 otherwise. We also interact ORIGIN with LNBRANCHMKT, to control for differential effects of the size of branches in the area where the bank is likely to have the highest reputation benefits. The definitions of variables and sample statistics are reported in Table 5.

4.1 Results

We employ bank-market observations referring to 103 provinces and 169 banks that employ jointly the traditional and the electronic delivery channels.¹⁰ The adoption of e-banking is negligible prior to 1998 so we employ data referring to the years 1999-2001. We include only the banks whose customers can perform transactions electronically, excluding accounts that give access to information only. We exclude cooperative banks because they are subject to limits in branching that force them to operate only in one or few provinces, which would bias the relationship between branches and e-banking. We exclude also the offices of foreign banks because their product mix is different from that of domestic banks.

The model is estimated with 2SLS with bank fixed effects so that any effect should be interpreted in terms of deviations from each bank's mean across markets. The results (Table 6) show that the number of branches has a non-monotonic effect on the diffusion of e-banking, described by a U-shaped curve. The minimum of the curve, where the elasticity changes sign, is reached at a value of 4.3 branches, which is lower than the sample mean but significantly above the median (1). Around 75% per cent of the observations lie below this value. The finding is consistent with the <u>substitution hypothesis</u> for the relevant range because banks appear to have more e-customers where they have a relatively lower number of branches. However, they cannot fully substitute branches. A positive effect is found for banks that have a very large number of

¹⁰ A larger sample of 259 banks includes also banks that do not supply e-banking services and pure internet banks and was employed for robustness purposes.

branches in certain markets. A possible explanation could be either that there are externalities in marketing or exogenous factors that apply to areas where there is very high density and which are not fully controlled for by urbanization and other characteristics. An interesting finding is the negative effect of the interaction term between LNBRANCHMKT and ORIGIN. The negative effect suggests that the benefit of opening a branch on e-customers is lower in the market where the bank was chartered. This result is likely to depend on the fact that reputation and network effects are marginally declining in branch density, since banks have a high number of branches in their "home" markets. The magnitude of the coefficient of the interaction term is large enough to almost compensate for the positive effect of the quadratic term, suggesting that the positive segment of the relationship between e-banking and branches is not relevant, given that most of the observations with high values of LNBRANCHMKT are those for which ORIGIN is equal to 1. The turning point when LNBRANCHMKT is set equal to 1 corresponds to a number of branches equal to 8.9.

Unexpectedly, the diffusion of e-banking is negatively affected by the share of households that have a PC; the sign of the coefficient is negative and statistically significant. An explanation of this result could be that PCs are measured at the same time as e-banking and endogeneity may produce a bias; in addition, convergence effects may be relevant, as we expect that PC ownership grows faster where initially lagged behind. As expected, e-banking is more developed where banks hold a larger volume of shares in their portfolio. POSPOP_1 is positive, as expected, but not significant. The level of education of the population, measured by EDUCATION, has a positive effect on e-banking, as predicted. E-banking is more present in less concentrated markets, suggesting that banks may promote the use of it as part of competitive strategies in retail markets. In addition, the result could be explained by the fact that highly concentrated markets in Italy are also less financially developed and consumers could be less prone to innovative services. Structural features such as SHURBAN and POPMOUNTAIN do not have a statistically significant effect. Finally, the dummy ORIGIN is positive, showing that on average the proportion of e-customers is higher in the bank's "home" market.

The results for equation (2) show that total accounts are positively correlated to ecustomers meaning that an increase in e-accounts implies an expansion of total customers, but with an elasticity less than unity. The number of branches that a bank has is strongly positively related to the total number of customers with checking accounts, as expected. The benefit of opening an additional branch is declining in magnitude but remains positive in the range observed in the data. The interaction term between ORIGIN and LNBRANCHMKT has a positive cpefficient, indicating that the elasticity of customers with respect to changes in branches is higher in the market where the bank was chartered, partly offsetting the negative quadratic term. Province value added has a negative sign, which may appear counterintuitive. The sign could be driven by the fact that the regression estimates the effect of value added, controlling for branches and for other financial variables (mortgages). In particular, the results indicate implicitly that the ratio of customers with checking accounts per branch is higher in areas with lower value added, which is consistent with the intuition that branches are probably fewer in lower income regions. The log of household mortgages is highly significant and positive, as expected. With the exception of EDUCATION, the other variables are not significant.

The robustness of the results is tested employing different samples. We estimated the model including the bank-market cells that correspond to zero branches, zero checking accounts and electronic accounts, under the assumption that banks can expand potentially in any market, either remotely or by opening a branch. The results are reported in Table 6-B and are consistent with our previous result, for both the main and the control variables. Finally, in Table 6-C we report the results of another specification, in which we included market fixed effects and removed the control variables that do not vary across time for each market. The motivation is that our main specification may exclude relevant differences across markets, causing omitted variable bias. The results are again consistent with our main estimation. The effect of branches on e-customers in equation (1) is non linear and suggests a dominance of the <u>substitution hypothesis</u> in the relevant range of the data. An interesting result is that PCHOUSE is no longer significant and positive, which provides support for the finding of a negative sign in the main specification, i.e. that without fixed effects the variable captures a mixture of levels and growth effects of the diffusion of computers. We prefer the main specification because with the fixed effects many of the instruments are no longer significant, weakening the identification of the model.

5. The Impact of Electronic Provision on Revenues

5.1 The Empirical Model: A Non-Standard Revenue Function

The goal of our second test is to measure indirectly the extent to which customers are willing to pay for the joint access to banking services electronically and at branches. Our approach is based on estimating the effect of electronic delivery on the revenues per customer. The effect of e-banking is measured by the coefficient of the variables measuring the number of e-banking customers (e-customers), controlling for the total number of bank customers. We hold constant the size of the customer base and output volumes, and study how the share of customers that can purchase the bank's services electronically affects revenues.

We start from a non-standard indirect revenue function following (Berger, Humphrey and Pulley, 1996) specifying bank revenues as a function of output and input quantities. This model is considered more appropriate than the standard revenue function because banks are assumed to adjust output prices in response to given output quantities and maximization is performed over output prices.¹¹ We modify the non-standard revenue function to take into account the different ways through which the bank sells its services. Revenues are a function of output and input quantities (or prices for robustness purposes), and of the relative importance of face-to-face and e-banking provision of services. Since we do not have data on the number of transactions performed face-to-face or electronically, we use the number of customers as a proxy of quantities sold.

Rather than considering total revenues, we model only the subset of bank revenues generated by those services that are sold also on line in Italy in the relevant period : i) brokerage and asset management services, ii) payment services, iii) checking accounts. These services are standardized and homogeneous, hence a single electronic transaction can be considered a

¹¹ The standard revenue function is based on the assumption of a price-taking firm choosing the optimal output quantities, given input quantities. This assumption is unlikely to hold for a large number of bank outputs because bilateral contracting in which banks have some bargaining power is a common feature of most intermediation services.

substitute to a branch transaction regardless the issue of complementarity of the entire bundle of banking services.

Information on the first two categories of revenues is directly available from bank income statements in the form of commissions and fees from brokerage, asset management and payment services. Revenues from asset management services include: net income generated by commissions charged on securities transactions, (flat) fees for holding a securities portfolio, transactions on mutual funds shares, and other private banking services.

The revenues generated by checking accounts are included because in the Italian banking system an e-banking account is tied to a checking account and checking accounts generate an implicit income for banks, on top of the fees for payment services and other commissions. The reason is that banks generally pay on these deposits below-market interest rates as a compensation for liquidity services provided to customers and at the same time invest at market rates the liquidity collected. We estimate such implicit return as the difference between the interest flow obtained by applying the money market rate to the volume of checking accounts and the actual interest flow paid by the bank. The share of these revenues on total banks' revenues for different types of institutions is reported in Table 4.

The estimation of the revenue function requires the definition of the outputs. Given that we model only a subset of revenues, we restrict the analysis to those bank outputs that generate these revenues. We resort to measures of the stock of financial assets that determine the transactions because no information on the actual number of transactions is available.

In the case of brokerage and asset management services, we employ as a proxy of the volume of transactions the stocks of assets that customers hold at the bank. Households hold securities at banks under two types of arrangements involving different fees and commissions and different volumes of transactions per customer. The first arrangement implies that the bank hold the securities on behalf of the customer without providing any asset management service (essentially, customers pay a flat fee for the account plus brokerage commissions when the bank places orders on behalf of the holder). We distinguish these securities into shares (SHARES) and other securities (OTHSEC) because the evolution of stock prices in the period considered has been a particularly important determinant of the growth of e-banking.

The second arrangement implies the provision of some portfolio management service. We define the volume of securities managed (SECMAN) as the sum of the value of mutual fund shares held by customers and the value of other assets managed by the bank on behalf of the customers (private banking services). Most mutual funds are owned by banks that place the shares through their branch network or through financial salesmen.

In the case of payment services and revenues generated by checking account transactions we consider two dimensions of output. The first is the yearly average of monthly holdings in checking accounts (DEPBK). The second is the number of checking accounts since there is a large component of revenues from fixed fees, independent of the number of transactions or the size of the account (TOTCBK).

We include three input quantities: the number of employees (STAFF), the number of branches (BRANCHBK), and the number of investment professionals (INVPROF), in natural logs. No financial inputs are considered because asset management and payment services do not modify the levels of bank assets or liabilities. Another relevant input as far as payment services are concerned is the number of Points of Sale (POS) where customers can use debit cards for purchases. In our main specification we prefer to include input quantities rather than prices because prices cannot be measured accurately.

Our key variable is the number of e-banking customers (ECUSTBK). We include also the number of customers with phone banking accounts (PHONEBK) because for some banks phonebanking could be not negligible. In both cases we count only the customers who can perform transactions and exclude connections that provide information only. The coefficient of E-CUSTBK (and of PHONEBK) measure the marginal effect on bank revenues of increasing the number of customers connected by computer, given the total number of customers and the average portfolio composition. A positive effect implies that letting a customer access the bank both by e-banking and at the branch increases the revenues the bank gets from servicing such customer, consistent with the complementarity hypothesis. Consumers are paying more for having the opportunity of buying the same bundle of services using a multiplicity of delivery channels. A negative effects implies that revenues per customer are lower than without connection. Since we control for output quantities, an implication of a negative effect is that the average price has to be lower when a customer uses both delivery channels This interpretation is consistent with the low marginal cost of the e-banking transactions and that competition is effective because e-banking and branches are substitute rather than complement in the delivery of banking services.

Given the limited number of observations, we choose a very parsimonious functional form.¹² Outputs are expressed in logs with a linear and a quadratic term. Input quantities and the number of customers are in logs with a linear term only.

The resulting equation is:

$$LNREV_{it} = \alpha + \beta_{1} \bullet LNSHARES_{it} + \beta_{2} \bullet LNSHARES_SQ_{it} + \beta_{3} \bullet LNOTHSEC_{it}$$

$$+ \beta_{4} \bullet LNOTHSEC_SQ_{it} + \beta_{5} \bullet LNSECMAN_{it} + \beta_{6} \bullet LNSECMAN_SQ_{it}$$

$$+ \beta_{7} \bullet LNDEPBK_{it} + \beta_{8} \bullet LNDEPBK_SQ_{it}$$

$$+ \delta_{1} \bullet LNSTAFF_{it} + \delta_{2} \bullet LNBRANCHBK_{it} + \delta_{3} \bullet LNINVPROF_{it} + \delta_{4} \bullet LNPOS_{it}$$

$$+ \gamma_{1} \bullet LNTOTCBK_{it} + \gamma_{2} \bullet LNE-CUSTBK_{it} + \gamma_{3} \bullet LNPHONEBK_{it} + \Phi_{i} + \mathbf{T}_{t}.$$
(3)

The vectors Φ and \mathbf{T} are bank and time specific effects. We employ bank-specific fixed effects to control for quality differences in the services provided without ruling out that unobservable bank-specific output characteristics are correlated with the regressors.

For robustness purposes, we estimate equation (3) also with a narrower definition of revenues, including only brokerage and asset management services. The motivation is that brokerage commissions are the largest component of revenues from remote transactions. Our main results are unchanged so we do not show the results.

¹² Unlike the more familiar case of cost and profit functions, there are no theoretical reasons for selecting a specific functional form for the (non-standard) revenue function. Estimates obtained with a translog functional form do not improve on the linear-quadratic specification so we show only the latter.

5.2 Results

We employ annual data on virtually all Italian banks from Supervisory Reports for the period 1998-2001 to estimate equation (3). A first estimation is conducted employing all available observations (full sample) referring to banks that have at least one branch, whether they are active or not in the e-banking business. We exclude pure Internet banks because they have no branches other than the headquarters. Descriptive statistics for the full sample are reported in Table 7.

In a second estimation we included only banks with a positive number of customers performing e-banking transactions (click-and-mortar sample). The reason for this regression is to control for potential endogeneity between the size of e-banking operations and revenues if the decision to supply e-banking services is systematically related to past revenues. Banks with lower revenues could have been more active in innovating their product mix to increase their revenues or, on the contrary banks with higher revenues could have been able to invest in the fixed cost of entering the e-banking business. By excluding the banks that have not introduced electronic facilities we study the impact of e-banking only for banks that have already made the decision to supply it.

Finally, we replicate the estimation employing a third sub-sample excluding the small community banks (cooperative banks), which usually have very limited geographical reach since they tend to have branches in a very small number of provinces. In addition, they are organized as mutual banks and are likely to have different incentives in deciding their product mix.

The results of the estimation for the three samples are reported in Table 8. In all regressions the coefficients of the output terms are estimated with little precision due to the usual multicollinearity problems, even if the signs and magnitudes are plausible. The coefficient of the number of customers is always positive and statistically different from zero, as expected since revenues increase with the number of customers with checking accounts. Expanding the customer base increases revenues even if the volumes of deposits and securities is fixed.

The coefficient of the number of customers with e-banking transaction accounts is small but statistically different from zero. This finding is robust across all three samples. For the full sample an increase in the number of customers using e-banking facilities by 2.5 times, which is the order of magnitude of the actual increase experienced in 2001, leads to a 1 per cent reduction of revenues, every other thing equal (Table 8, column i). The effect is larger when the sample is limited to "click-and-mortar" banks (Table 8, column ii). In all regressions the coefficient of PHONEBK is negative but not statistically significant. The sign suggests that an increase in the number of customers with phone banking tends to reduce revenues, although the effect is estimated with little precision because the variable takes values that are very close to zero.

For robustness purposes, we estimated the revenue function replacing variable input quantities with variable input prices for all the three samples. We included two variable input prices: the unit cost of labour (PLABOR) and the ratio of other operating costs (non-interest costs excluding labour expenses) to total assets (POTHER). The results are consistent with our previous findings and the negative impact of e-banking on unit revenues is robust (Table 8, columns iv-vi).

The analysis of the non-standard revenue function suggest that the provision of e-banking services has produced a negative effect on per customer revenues. Given the structure of the revenue function, this effect is driven by the lower prices charged to services delivered electronically because the levels of financial assets and the total number of bank customers are held constant. The finding suggests that customers have substituted within the same bank branch transactions with e-banking transactions and that the lower prices charged for e-banking transactions have not been compensated by a large enough increase in the overall number of transactions. This interpretation is consistent with information on the prices charged by "click and mortar banks" for e-banking services, based on case studies and occasional surveys. The survey conducted by the Bank of Italy on Internet banking in Italy indicates that 40 per cent of the banks offering e-banking charge lower prices on all types of on-line transactions than on the equivalent branch transactions. One half of the banks surveyed reported charging lower prices on some of their on-line services. Only one tenth responded that they charged the same price for face-to-face and the corresponding Internet transactions.

A caveat that should be kept in mind in interpreting the results is that the sample period covers the take-off of Internet banking and this may be reflected in very aggressive pricing strategies that do not represent equilibrium conditions. Although we do not have evidence

consistent with this view, in the Bank of Italy survey for 2001 banks reported that marketing financial products with new characteristics was the primary objective of their e-banking strategies.

6. Conclusion

In this study we investigated the existence of complementarities between the physical delivery of banking services at traditional branches and the provision of on-line banking services, employing a new data-set on Italian banks' operations in a large set of local markets. We test the widely held view that electronic delivery is a complement to branches and that the "click and mortar" banking model is in a better position to extract the consumer surplus generated by the new technologies. Due to the lack of data on prices and quantities of services delivered through alternative channels by each bank, we resorted to two indirect testing procedures.

We first analyzed the pattern of diffusion of e-banking customers of "click and mortar" banks across local markets to test three alternative hypotheses. The <u>complements hypothesis</u> predicts a positive relationship between branches and the number of e-customers, once the total number of customers is controlled for, meaning that consumers value the size of the branch network when they make a decision about establishing an electronic connection to a bank. The <u>substitution hypothesis</u> predicts a negative effect of branches on the diffusion of e-banking, meaning that consumers choose more frequently to establish an electronic connection with a bank when that bank has a smaller branch network in the relevant local market. Ceteris paribus, the preference for e-banking is driven by transaction costs. The <u>segmentation hypothesis</u> states that the two bundles of services - traditional and electronic - are perceived as independent by consumers, whose choices on e-banking are unrelated to the quality of the branch network.

Our results are not consistent with complementarity between branch and electronic access to banking services, suggesting the dominance of the <u>substitution hypothesis</u> for most of the range of the data in terms of branches. Consumers tend to demand, ceteris paribus, more ebanking connections from banks that have a smaller number of branches. A different regime appears to hold in markets where banks have a disproportionately large branch network since the elasticity of e-banking to branches turns positive, possibly because of reputation effects.

In the second part of the analysis we explored indirectly the impact of e-banking on the revenues that banks can extract from each consumer, focusing exclusively on the revenues generated by brokerage and asset management services and payments, which are the services that Italian banks provide electronically. Our results suggest that a high share of e-customers, given the size of the customer base, is associated with a fall in revenues per customer, suggesting that banks did not extract surplus from the joint provision of traditional and electronic services in the period examined.

The two results are consistent with two possible explanations. The first is that e-banking is not strictly characterized by complementarities with traditional banking and that a substantial component of the advantage of the "click and mortar" model depends on the initial customer base of these banks. A second explanation is that past pricing policies have reflected strategic motives and that prices reflected disequilibrium conditions in the industry. In the first scenario e-banking is viable as long as cost reductions follow the expansion of e-accounts and/or higher value added services are supplied. Such scenario is consistent with the overall reassessment of the "click and mortar" model under way in the banking industry, several large European banks having announced major restructuring plans to downsize their branch networks and enhance their ebanking facilities in terms of quality and tailoring of services. Tables

Table 1 Number of banks offering e-banking and phone banking services to households

Number of banks offering services through personal computer connections (e-banking) or phone calls (phone-banking). Transactional services are those in which the customer is allowed to use the remote connection to send orders that will be executed by the banks. Large banks are those with total assets greater or equal to 20 billion euros, medium-sized banks are those with total assets between 7 and 20 billion euros. Branches of foreign banks are excluded. Figures are end-of-period data.

	1998	1999	2000	2001
Large banks				
e-banking services	6	8	10	13
-transactional	5	6	10	13
phone banking services	7	8	9	9
-transactional	7	7	8	8
Total number of large banks	16	16	18	18
Medium-sized banks offering				
e-banking services	11	13	14	17
-transactional	11	12	14	17
phone banking services	6	8	6	9
-transactional	3	4	4	6
Total number of medium-sized banks	27	29	26	27
Credit cooperative banks offering				
e-banking services	91	190	299	341
-transactional	82	176	290	334
phone banking services	78	81	107	96
-transactional	2	5	15	14
Total number of credit cooperative banks	566	532	498	474
Other small hanks offering				
e-banking services	61	73	118	143
-transactional	53	66	115	138
phone banking services	28	35	51	50
-transactional	7	12	22	28
Total number of other small banks	246	234	231	240
Total hanks				
e-banking services	169	284	441	514
-transactional	151	260	429	502
phone banking services	119	132	173	164
-transactional	19	28	49	56
Total number of banks	855	811	773	759

Table 2Share of e-banking and phone banking customers by type of bank

Number of customers belonging to the household sector using e-banking or phone-banking services as a percentage of the total number of bank depositors. Data are collected at the bank level so an individual with relationships with more than one bank is counted as many times as the number of relations. Transactional services are those in which the customer is allowed to use the remote connection to send orders that will be executed by the banks. Large banks are those with total assets greater or equal to 20 billion euros, medium-sized banks are those with total assets between 7 and 20 billion euros. Branches of foreign banks are excluded. Figures are end-of-period data.

	1998	1999	2000	2001
Large banks				
e-banking services	0.12	0.76	2.06	6.48
-transactional	0.11	0.72	1.66	5.25
phone banking services	3.04	4.06	7.58	7.12
-transactional	0.03	0.03	0.07	0.07
Medium-sized banks				
e-banking services	0.32	0.49	2.42	8.35
-transactional	0.20	0.28	2.06	6.87
phone banking services	1.17	3.75	2.86	1.75
-transactional	0.01	0.01	0.01	0.01
Credit cooperative banks				
e-banking services	0.08	0.39	1.65	2.85
-transactional	0.06	0.32	1.35	2.51
phone banking services	5.38	5.45	5.73	5.61
-transactional	0.00	0.00	0.00	0.00
Other small banks				
e-banking services	0.52	0.39	6.08	12.82
-transactional	0.41	0.33	5.49	11.31
phone banking services	8.03	10.50	10.85	14.67
-transactional	0.01	0.04	0.07	0.12

Table 3Share of e-banking and phone banking customers by geographical area

Number of customers belonging to the household sector using e-banking or phone-banking services as a percentage of the total number of bank depositors. Data are collected at the bank level so an individual with relationships with more than one bank is counted as many times as the number of relations. Figures are end-of-period data and do not include those reported by branches of foreign banks.

	1998	1999	2000	2001
Northern regions				
e-banking services	0.36	0.57	3.12	8.38
-transactional	0.26	0.46	2.47	6.92
phone banking services	5.18	6.71	7.15	7.55
-transactional	0.02	0.03	0.05	0.06
Central regions				
e-banking services	0.24	0.78	3.59	10.20
-transactional	0.21	0.73	3.22	8.78
phone banking services	3.47	4.92	7.08	8.25
-transactional	0.02	0.03	0.05	0.07
Southern regions				
e-banking services	0.06	0.46	2.17	6.36
-transactional	0.05	0.41	1.85	5.60
phone banking services	2.19	3.57	5 92	7 77
-transactional	0.02	0.02	0.05	0.07
Italy				
e-banking services	0.26	0.59	2.97	8.22
-transactional	0.20	0.50	2.46	6.95
nhone hanking services	4.05	5 55	6.81	7 75
_transactional	-1.05	0.03	0.01	0.06
-u ansactional	0:02	0.05	0.05	0.00

Table 4Composition of bank revenues

Interest revenues on checking accounts are estimated as the product of the stock of checking deposits and the 3-month money market rate minus the actual interest payments made by the bank to the depositors. Revenues from payment services and from investment and management services are obtained as the difference between fees and commissions receivable and fees and commission payable. Large banks are those with total assets greater or equal to 20 billion euros. medium-sized banks are those with total assets between 7 and 20 billion euros. Branches of foreign banks are excluded. Figures are end-of-period data.

	1998	1999	2000	2001
Large banks				
Net interest income	58.9	54.2	49.1	50.4
- checking accounts (estimate)	16.0	12.6	16.4	13.5
Non interest income	41.1	45.8	50.9	49.6
-payment services	3.7	4.0	4.0	4.6
-investment and asset management	13.1	16.3	16.0	11.5
Gross income	100.0	100.0	100.0	100.0
Medium-sized banks				
Net interest income	58.9	55.6	51.4	57.9
- checking accounts (estimate)	14.4	11.0	17.0	14.9
Non interest income	41.1	44.4	48.6	48.1
-payment services	5.2	5.6	4.8	5.1
-investment and asset management services	11.2	14.8	18.3	12.3
Gross income	100.0	100.0	100.0	100.0
Credit cooperative banks				
Net interest income	72.8	74.4	76.1	78.2
- checking accounts (estimate)	12.0	9.6	16.7	15.1
Non interest income	27.2	25.6	24.0	21.8
-payment services	4.6	5.5	5.9	6.3
-investment and asset management	4.1	5.9	7.9	5.0
Gross income	100.0	100.0	100.0	100.0
Other small banks				
Net interest income	62.4	60.3	58.2	59.5
- checking accounts (estimate)	12.8	9.6	15.1	13.2
Non interest income	37.6	39.7	41.8	40.5
-payment services	4.5	5.0	5.4	6.0
-investment and asset management	9.4	11.5	13.3	9.6
Gross income	100.0	100.0	100.0	100.0
Total banks				
Net interest income	60.6	57.0	52.8	55.2
- checking accounts (estimate)	14.7	11.5	16.3	13.8
Non interest income	39.4	43.0	47.2	44.8
-payment	4.3	4.8	4.5	5.0
-investment and asset management	11.2	14.4	15.5	19.9
Gross income	100.0	100.0	100.0	100.0

Table 5Definitions of variables and descriptive statistics

The sample includes only banks that were supplying both e-banking and traditional services. Cooperative banks are excluded. Each observation refers to bank-market-year.

Symbol		Mean	Min	Max	S. dev.		
Dependent Variables							
LNE-CUSTMKT	Natural log of the number of customers with electronic access of bank i in market j at time t	2.715	0	10.822	2.298		
E-CUSTMKT	Number of customers with electronic access of bank i in market j at time t.	331.4	0	50104	1786.1		
LNTOTCMKT	Natural log of the total number of checking accounts of bank i in market j at time t.	4.205	0	13.554	4.214		
ТОТСМКТ	Total number of checking accounts of bank i in market j at time t.	6880.9	0	769659	26543.8		
	Explanatory Variables						
LNMORTGAGE	Natural log of the volume of household mortgages of bank i in market j	1.533	0	7.789	1.651		
LNBRANCHMKT	Log of the number of branches of bank i in market j.	0.947	0	5.583	1.204		
HERFDEP_1	Provincial Herfindahl index of deposit concentration at the end of previous year.	0.165	0.049	0.527	0.070		
POSPOP_1	Number of Points of Sale of market j at time t-1 divided by population (in 1.000)	9.147	0.738	36.248	5.029		
LNSHAREMKT_1	Log of the volume of shares held in households' portfolios at banks in market j, end-of previous year stocks	7.864	0	17.492	3.790		
PCHOUSE	Share of households with a PC (regional)	29.64	13.6	42.1	7.231		
LNVADDED	Natural log of per capita province value added (1,000 euro).	9.101	7.138	11.627	0.869		
LNPOP	Natural log of population in the province (1,000).						
SHURBAN	Share of population living in the main city of the province (1998).	28.545	8.642	15.357	87.638		
ROADS	Kilometers of roads to the area in 1,000 square kilometers, regional.	28.87	15.85	42.82	4.89		
POPMOUNTAIN	Share of the population living in towns located at more than 600 meters above the sea level (1998).	0.064	0	0.797	0.128		
EDUCATION	Natural log of the number of people with a high school diploma or higher (1998).	37.29	22.40	49.459	4.183		
ORIGIN	Equal to 1 if bank i is chartered in market j, 0 otherwise.	0.046	0	1	0.209		
BRANCHMKT	Number of branches of bank i in market j	6.063	0	265	16.911		
N. of banks	169						
N. observations	8023						

Table 6-ABank-market test, main results

Bank dummy variables are included. Estimated with 2SLS; robust standard errors below coefficients. Standard errors are corrected for clustering at the local market level. The panel is unbalanced. Bank-market cells with no customers and no branches are dropped.

Equation (1)		Equation (2)			
Explanatory Variables	LNE-CUSTMKT	Explanatory Variables	LNTOTCMKT		
LNTOTCMKT	0.406 ***	LNE-CUSTMKT	-0.022		
	0.022		0.348		
LNBRANCHMKT	-0.790 ***	LNBRANCHMKT	4.436 ***		
	0.130		0.352		
LNBRANCHMKT SQ	0.273 ***	LNBRANCHMKT SQ	-0.828 ***		
	0.022		0.045		
LNVADDED	-0.014 *	LNVADDED	0.013 **		
	0.008		0.006		
EDUCATION	0.005	EDUCATION	-0.002		
	0.004		0.004		
PCHOUSE	-0.002	LNPOP	-0.093		
	0.005		0.059		
POSPOP_1	0.010 ***	LNMORTGAGE	1.121 ***		
	0.003		0.164		
LNSHAREMKT_1	0.121 ***	SHURBAN	0.001		
	0.019		0.001		
SHURBAN	-0.000	HERFDEP_1	-0.039		
	0.001		0.237		
POPMOUNTAIN	0.025	ORIGIN	-1.241		
	0.074		1.190		
ROADS	0.003	ORIGIN*LNBRANCHMKT	0.661 *		
	0.003		0.338		
HERFDEP_1	-0.351 *	DU99	-0.124		
	0.191		0.664		
ORIGIN	1.318 ***	DU00	0.168		
	0.299		0.296		
ORIGIN*LNBRANCHMKT	-0.406 ***	CONSTANT	0.685 **		
	0.082		0.300		
DU99	-1.812 ***				
	0.087				
DU00	-0.959 ***				
	0.046				
CONSTANT	0.317				
	0.242				
Adjusted R-squared	0.775	Adjusted R-squared	0.934		
F(14, 7840)	1040.5	F(15, 7839)	3276.6		
Number of obs.	8023	Number of obs.	8023		

Table 6-B Bank-market Test, no Restrictions on expansion

Bank dummy variables are included. Estimated with 2SLS; robust standard errors below coefficients. Standard errors are corrected for clustering at the local market level. The panel is unbalanced. Bank-market cells with no customers and no branches are included.

Equation (1)		Equation (2)			
Explanatory Variables	LNE-CUSTMKT	Explanatory Variables	LNTOTCMKT		
LNTOTCMKT	0.457 ***	LNE-CUSTMKT	-0.096		
	0.023		0.191		
LNBRANCHMKT	-1.004 ***	LNBRANCHMKT	4.914 ***		
	0.148		0.217		
LNBRANCHMKT_SQ	0.316 ***	LNBRANCHMKT_SQ	-0.907 ***		
	0.026		0.041		
LNVADDED	-0.007 *	LNVADDED	0.003 *		
	0.004		0.002		
EDUCATION	0.005 **	EDUCATION	-0.002		
	0.002		0.001		
PCHOUSE	0.001	LNPOP	-0.018		
	0.002		0.019		
POSPOP_1	0.003 *	LNMORTGAGE	1.058 ***		
	0.002		0.101		
LNSHAREMKT_1	0.063 ***	SHURBAN	-0.000		
	0.007		0.001		
SHURBAN	0.001	HERFDEP_1	0.013		
	0.001		0.067		
POPMOUNTAIN	0.021	ORIGIN	-0.961		
	0.034		1.066		
ROADS	-0.001	ORIGIN*LNBRANCHMKT	0.579 *		
	0.001		0.319		
HERFDEP_1	-0.157 *	DU99	-0.079		
	0.084		0.106		
ORIGIN	0.814 *	DU00	-0.003		
	0.433		0.060		
ORIGIN*LNBRANCHMKT	-0.343 ***	CONSTANT	0.178 **		
	0.116		0.090		
DU99	-0.511 ***				
	0.039				
DU00	-0.332 ***				
	0.023				
CONSTANT	-0.287				
	0.133				
Adjusted R-squared	0.805	Adjusted R-squared	0.950		
F(14, 7840)	818.2	F(12,28272)	4788.7		
Number of obs.	28453	Number of obs.	28453		

Table 6-C Bank-market test, Model with market fixed effects

Bank dummy variables and market fixed effects are included (coefficients are not shown). Estimated with 2SLS; robust standard errors below coefficients. Standard errors are corrected for clustering at the local market level. The panel is unbalanced. Bank-market cells with no customers and no branches are dropped.

Equation (1)		Equation (2)			
Explanatory Variables	LNE-CUSTMKT	Explanatory Variables	LNTOTCMKT		
LNTOTCMKT	0.393 ***	LNE-CUSTMKT	-1.635 **		
	0.022		0.812		
LNBRANCHMKT	-0.662 ***	LNBRANCHMKT	5.985 ***		
	0.131		0.814		
LNBRANCHMKT SQ	0.246 ***	LNBRANCHMKT SQ	-0.942 ***		
	0.023		0.065		
LNVADDED	0.117 **	LNVADDED	0.033		
	0.049		0.071		
PCHOUSE	0.009	LNPOP	4.850		
	0.008		4.618		
POSPOP_1	0.015	LNMORTGAGE	1.840 ***		
	0.014		0.363		
LNSHAREMKT_1	-0.041	HERFDEP_1	-0.056		
	0.049		1.155		
HERFDEP_1	-0.950	ORIGIN	-0.087		
	1.417		1.371		
ORIGIN	1.157 ***	ORIGIN*LNBRANCHMKT	0.504		
	0.309		0.357		
ORIGIN*LNBRANCHMKT	-0.339 ***	DU99	-3.241 **		
	0.085		1.492		
DU99	-1.584 ***	DU00	-1.211 *		
	0.162		0.646		
DU00	-0.702 ***	CONSTANT	-34.352		
	0.098		35.097		
CONSTANT	-0.437				
	1.277				
Adjusted R-squared	0.776	Adjusted R-squared	0.936		
<i>F(101, 7741)</i>	10336.7	F(15, 7839)	92113.9		
Number of obs.	8023	Number of obs.	8023		

Table 7 Revenue analysis: definitions of variables and descriptive statistics

All financial variables are measured in thousands of euros. Statistics refer to an unbalanced panel of 809 banks between 1998 and 2001. The number of observations is 2799.

Symbol		Mean	Min	Max	S. dev.	
Dependent Variable						
LNREV	Natural log of the revenues from investment and asset management services, payment services and deposits in checking accounts.	7.699	2.447	14.497	1.891	
	Explanatory Variables					
LNOTHSEC	Natural log of the volume of securities held in custody at the bank by customers; shares excluded.	11.084	0	18.606	2.491	
LNSHARES	Natural log of the volume of shares held in custody at the bank by customers.	7.275	0	16.103	3.338	
LNSECMAN	Natural log of the volume of mutual fund shares and other securities for which the bank provides asset management services	4.907	0	17.943	5.500	
LNDEPBK	Natural log of the volume of deposits in checking accounts	10.867	5.482	17.536	1.771	
LNSTAFF I NBRANCHBK	Natural log of the number of staff.	4.161	1.098	10.444	1.633	
LNINVPROF	Natural log of the number of investment professionals.	0.328	0	9.216	1.040	
LNPOS	Natural log of the number of "points of sale".	3.579	0	11.515	2.612	
LNPLABOR	Natural log of the per unit labor expenses.	4.011	2.823	6.528	0.163	
LNPOTHER	Natural log of the ratio of non-interest non-staff expenses to total assets.	-4.240	-6.797	-0.349	0.317	
LNTOTCBK	Natural log of the number of customers with a checking account at the bank.	8.823	0.693	15.020	1.690	
LNE-CUSTBK	Natural log of the number of customers with an account enabled for transactions through a PC connection.	2.036	0	12.955	2.634	
LNPHONEBK	Natural log of the number of customers with an account enabled for transactions by phone.	0.347	0	12.583	1.704	

Table 8Non-standard revenue function

Regressions are estimated on an unbalanced panel of individual yearly bank data over the period between 1998 and 2001. Joint producers are banks that are active in both the traditional and remote banking business. The restricted sample is obtained excluding community banks from the full sample.

Dependent variable:	Specification: Input quantities			Specification: Input prices		
LNREV						
	Full	Joint	Restricted	Full	Joint	Restricted
	sample	producers	sample	sample	producers	sample
	(i)	(ii)	(iii)	(iv)	(v)	(vi)
LNSHARES	-0.004	-0.013	-0.022	-0.004	-0.003	-0.022
	0.006	0.020	0.014	0.006	0.0194	0.014
LNSHARES_SQ	0.001	0.001	0.002	0.001	0.001	0.002
	0.001	0.001	0.001	0.001	0.001	0.001
LNOTHSEC	-0.029 **	0.012	0.061	-0.041 ***	0.019	0.025
	0.014	0.042	0.064	0.014	0.040	0.065
LNOTHSEC_SQ	0.004 ***	0.003	-0.002	0.005 ***	0.002	0.001
	0.001	0.003	0.003	0.001	0.002	0.003
LNSECMAN	-0.001	-0.001	-0.002	-0.001	-0.007	-0.003
	0.004	0.012	0.005	0.004	0.012	0.005
LNSECMAN_SQ	0.001 *	0.001	0.001 *	0.001 *	0.001	0.002 **
	0.000	0.001	0.001	0.000	0.001	0.001
LNDEPBK	0.747 ***	0.506 ***	0.697 ***	0.763 ***	0.348 ***	0.690 ***
	0.075	0.121	0.097	0.078	0.126	0.108
LNDEPBK_SQ	-0.012 ***	-0.001	-0.003	-0.010 ***	0.007	0.001
	0.004	0.006	0.005	0.004	0.006	0.005
LNTOTCBK	0.128 ***	0.150 ***	0.038 **	0.150 ***	0.136 ***	0.075 **
	0.019	0.028	0.031	0.018	0.025	0.030
LNE-CUSTBK	-0.004 *	-0.007 *	-0.014 ***	-0.004 *	-0.008 **	-0.014 ***
	0.002	0.004	0.004	0.002	0.004	0.004
LNPHONEBK	-0.007 **	-0.006	-0.002	-0.007 **	-0.002	-0.003
	0.003	0.005	0.004	0.003	0.006	0.004
LNSTAFF	0.093 **	-0.065	0.182 **	-	-	-
	0.038	0.069	0.055	-	-	-
LNBRANCHBK	0.108 ***	0.146 **	0.072	-	-	-
	0.035	0.060	0.049	_	-	_
LNINVPROF	-0.002	-0.001	-0.004	_	-	_
	0.006	0.009	0.007	-	_	_
LNPOS	0.008 *	0.016 **	0.012 **	_	_	_
	0.004	0.007	0.006	-	_	_
LNPLABOR	_	_	_	0.220 ***	0.503 ***	0.010
	_	_	_	0.039	0.071	0.054
LNPOTHER	-	_	-	-0.034	-0.264 ***	0.012
	-	-	-	0.031	0.051	0.049
Adi, R ²	0.996	0.994	0.998	0.996	0.995	0.998
N. of obs.	2799	878	1312	2799	878	1312
N. of banks	809	274	529	809	274	529

References

- Allen, F., J. McAndrews and P.E. Strahan, 2002. "E-Finance: An Introduction", *Journal of Financial Service Research*, 22 (1-2): 5-25.
- Banca d'Italia, 2001a. "Indagine sull'impiego della rete Internet nel sistema bancario e finanziario italiano (A Survey on the diffusion of Internet within the Italian Banking and Financial System)", *Bollettino di Vigilanza* (Supervision Bulletin), July.
- Banca d'Italia, 2001b. Annual Report for the Year 2000, English version.
- Bank for International Settlements, 2003. *Payment and settlement systems in selected countries*, Basle.
- Berger, A.N., 2003. "The Economic Effect of Technological Progress: Evidence from the Banking Industry", *Journal of Money, Credit and Banking*, 35: 141-76.
- Berger, A.N., D.B. Humphrey and L.B. Pulley, 1996. "Do Consumers Pay for One-Stop Banking? Evidence from an Alternative Revenue Function", *Journal of Banking and Finance*, 20 (9): 1601-21.
- Berlin, M., L.J. Mester, 1999. "Deposits and Relationship Lending", *Review of Financial Studies*, 12 (3): 579-607.
- Caskey, J., 1994. "Who Has a Bank Account and Who Doesn't: 1977 and 1989," *Eastern Economic Journal*, 20 (1), Winter.
- Christiansen, H., 2001. "Electronic Finance: Economics and Institutional Factors", OECD, Financial Affairs Division, Occasional Papers, n. 2.
- Claessens, S., T. Glaessner and D. Klingebiel, 2002. "Electronic Finance: Reshaping the Financial Landscape around the World", *Journal of Financial Service Research*, 22 (1-2): 29-61.
- Clemons, E. K. and L.M. Hitt, 2001. "The Internet and the Future of Financial Services: Transparency, Differential Pricing, and Disintermediation", in Brookings Task Force on the Internet, *The Economic Payoff from the Internet Revolution*. Washington, DC: Brookings Institution Press.
- De Young, R., 2001. "The Financial Performance of Pure Play Internet Banks", Federal Reserve Bank of Chicago, *Economic Perspectives* 25 (1): 60-75.
- De Young, R. and W. C. Hunter, 2002. "Deregulation, the Internet, and the Competitive Viability of Large Banks and Community Banks", in B. Gulp (ed.), *The Future of Banking*, Greenwood Press.
- Dick, A. A., 2003. "Nationwide Branching and its Impact on Market Structure, Quality and Market Structure", Federal Reserve Board, Finance and Economics Discussion Series 2003-35.

- European Central Bank (ECB), 1999. The Effects of Technology on the EU Banking Systems. Frankfurt.
- European Central Bank (ECB), 2002. Structural Analysis of the EU Banking Sector. Frankfurt.
- Evanoff, D.D., 1988. "Branch Banking and Service Accessibility", *Journal of Money Credit and Banking*, 20 (2): 191-202.
- Furst, K., W.L. Lang and D.E. Nolle, 2001. "Internet Banking", Journal of Financial Service Research, 22 (1-2): 95-117.
- Group of Ten, 2001. *Report on the Consolidation in the Financial Sector*, Bank of International Settlements, Basle.
- JP Morgan Securities Ltd., 2000. Online Finance Europe. London.
- Kashyap, A., R. Rajan, and J. Stein, 2002. "Banks as Liquidity Providers: An Explanation for the Co-Existence of Lending and Deposit-Taking", *Journal of Finance*, 57 (1).
- Leamer, E. E. and P.E. Storper, 2001. "The Economic Geography of The Internet Age", NBER Working Paper n. 8450.
- Mishkin, F.S. and P.E. Strahan, 1999. "What Will Technology Do to Financial Structure?," in R. E. Litan and Santomero A. M. (eds.), Brookings – Wharton Papers on Financial Services: 1999.
- Sato, S. and J. Hawkins, 2001. "Electronic Finance: An Overview of the Issues", BIS Papers n. 7: 1-12.
- Stigler, G., 1961. "The Economics of Information", *Journal of Political Economy*, 69 (3): 651-659.
- Sullivan, R.J., 2000. "How Has the Adoption of Internet Banking Affected Performance and Risk in Banks?", *Financial Industry Perspectives*, Federal Reserve Bank of Kansas City, December: 1-16.
- Sullivan, R.J., 2001. "Performance and Operation of Commercial Bank Web Sites", *Financial Industry Perspectives*, Federal Reserve Bank of Kansas City: 23-33.

- N. 482 La distribuzione del reddito e della ricchezza nelle regioni italiane, by L. CANNARI and G. D'ALESSIO (June 2003).
- N. 483 *Risk aversion, wealth and background risk*, by L. GUISO and M. PAIELLA (September 2003).
- N. 484 What is this thing called confidence? A comparative analysis of consumer indices in eight major countries, by R. GOLINELLI and G. PARIGI (September 2003).
- N. 485 L'utilizzo degli indicatori compositi nell'analisi congiunturale territoriale: un'applicazione all'economia del Veneto, by P. CHIADES, M. GALLO and A. VENTURINI (September 2003).
- N. 486 Bank capital and lending behaviour: empirical evidence for Italy, by L. GAMBACORTA and P. E. MISTRULLI (September 2003).
- N. 487 A polarization of polarization? The distribution of inequality 1970-1996, by C. BIANCOTTI (March 2004).
- N. 488 Pitfalls of monetary policy under incomplete information: imprecise indicators and real indeterminacy, by E. GAIOTTI (March 2004).
- N. 489 Information technology and productivity changes in the banking industry, by L. CASOLARO and G. GOBBI (March 2004).
- N. 490 *La bilancia dei pagamenti di parte corrente Nord-Sud (1998-2000)*, by L. CANNARI and S. CHIRI (March 2004).
- N. 491 Investire in Italia? Risultati di una recente indagine empirica, by M. COMMITTERI (March 2004).
- N. 492 Centralization of wage bargaining and the unemployment rate: revisiting the hump-shape hypothesis, by L. FORNI (June 2004).
- N. 493 Endogenous monetary policy with unobserved potential output, by A. CUKIERMAN and F. LIPPI (June 2004).
- N. 494 Il credito commerciale: problemi e teorie, by M. OMICCIOLI (June 2004).
- N. 495 *Condizioni di credito commerciale e differenziazione della clientela*, byL. CANNARI, S. CHIRI AND M. OMICCIOLI (June 2004).
- N. 496 Il debito commerciale in Italia: quanto contano le motivazioni finanziarie?, by P. FINALDI RUSSO and L. LEVA (June 2004).
- N. 497 Funzionamento della giustizia civile e struttura finanziaria delle imprese: il ruolo del credito commerciale, by A. CARMIGNANI (June 2004).
- N. 498 Does trade credit substitute for bank credit?, by G. DE BLASIO (June 2004).
- N. 499 *Monetary policy and the transition to rational expectations,* by G. FERRERO (June 2004).
- N. 500 Turning-point indicators from business surveys: real-time detection for the euro area and its major member countries, by A. BAFFIGI and A. BASSANETTI (June 2004).
- N. 501 *La ricchezza delle famiglie italiane e americane*, by I. FAIELLA and A. NERI (June 2004).
- N. 502 Optimal duplication of effort in advocacy systems, by G. PALUMBO (June 2004).
- N. 503 Il pilastro privato del sistema previdenziale. Il caso del Regno Unito, by F. SPADAFORA (June 2004).
- N. 504 Firm size distribution and employment protection legislation in Italy, by F. SCHIVARDI and R. TORRINI (June 2004).
- N. 505 Social mobility and endogenous cycles in redistribution, by F. ZOLLINO (July 2004).
- N. 506 Estimating expectations of shocks using option prices, by A. DI CESARE (July 2004).
- N. 507 Estimating state price densities by Hermite polynomials: theory and application to the Italian derivatives market, by P. GUASONI (July 2004).

^(*) Requests for copies should be sent to:

Banca d'Italia – Servizio Studi – Divisione Biblioteca e pubblicazioni – Via Nazionale, 91 – 00184 Rome (fax 0039 06 47922059). They area available on the Internet www.bancaditalia.it.

1999

- L. GUISO and G. PARIGI, *Investment and demand uncertainty*, Quarterly Journal of Economics, Vol. 114 (1), pp. 185-228, **TD No. 289 (November 1996)**.
- A. F. POZZOLO, Gli effetti della liberalizzazione valutaria sulle transazioni finanziarie dell'Italia con l'estero, Rivista di Politica Economica, Vol. 89 (3), pp. 45-76, TD No. 296 (February 1997).
- A. CUKIERMAN and F. LIPPI, Central bank independence, centralization of wage bargaining, inflation and unemployment: theory and evidence, European Economic Review, Vol. 43 (7), pp. 1395-1434, TD No. 332 (April 1998).
- P. CASELLI and R. RINALDI, La politica fiscale nei paesi dell'Unione europea negli anni novanta, Studi e note di economia, (1), pp. 71-109, TD No. 334 (July 1998).
- A. BRANDOLINI, The distribution of personal income in post-war Italy: Source description, data quality, and the time pattern of income inequality, Giornale degli economisti e Annali di economia, Vol. 58 (2), pp. 183-239, TD No. 350 (April 1999).
- L. GUISO, A. K. KASHYAP, F. PANETTA and D. TERLIZZESE, Will a common European monetary policy have asymmetric effects?, Economic Perspectives, Federal Reserve Bank of Chicago, Vol. 23 (4), pp. 56-75, TD No. 384 (October 2000).

2000

- P. ANGELINI, Are banks risk-averse? Timing of the operations in the interbank market, Journal of Money, Credit and Banking, Vol. 32 (1), pp. 54-73, **TD No. 266 (April 1996).**
- F. DRUDI and R: GIORDANO, *Default Risk and optimal debt management,* Journal of Banking and Finance, Vol. 24 (6), pp. 861-892, **TD No. 278 (September 1996)**.
- F. DRUDI and R. GIORDANO, *Wage indexation, employment and inflation,* Scandinavian Journal of Economics, Vol. 102 (4), pp. 645-668, **TD No. 292 (December 1996)**.
- F. DRUDI and A. PRATI, *Signaling fiscal regime sustainability*, European Economic Review, Vol. 44 (10), pp. 1897-1930, **TD No. 335 (September 1998)**.
- F. FORNARI and R. VIOLI, *The probability density function of interest rates implied in the price of options,* in: R. Violi, (ed.), Mercati dei derivati, controllo monetario e stabilità finanziaria, Il Mulino, Bologna, **TD No. 339 (October 1998)**.
- D. J. MARCHETTI and G. PARIGI, Energy consumption, survey data and the prediction of industrial production in Italy, Journal of Forecasting, Vol. 19 (5), pp. 419-440, TD No. 342 (December 1998).
- A. BAFFIGI, M. PAGNINI and F. QUINTILIANI, Localismo bancario e distretti industriali: assetto dei mercati del credito e finanziamento degli investimenti, in: L.F. Signorini (ed.), Lo sviluppo locale: un'indagine della Banca d'Italia sui distretti industriali, Donzelli, TD No. 347 (March 1999).
- A. SCALIA and V. VACCA, *Does market transparency matter? A case study*, in: Market Liquidity: Research Findings and Selected Policy Implications, Basel, Bank for International Settlements, **TD No. 359** (October 1999).
- F. SCHIVARDI, *Rigidità nel mercato del lavoro, disoccupazione e crescita*, Giornale degli economisti e Annali di economia, Vol. 59 (1), pp. 117-143, **TD No. 364 (December 1999)**.
- G. BODO, R. GOLINELLI and G. PARIGI, *Forecasting industrial production in the euro area*, Empirical Economics, Vol. 25 (4), pp. 541-561, **TD No. 370 (March 2000)**.
- F. ALTISSIMO, D. J. MARCHETTI and G. P. ONETO, *The Italian business cycle: Coincident and leading indicators and some stylized facts*, Giornale degli economisti e Annali di economia, Vol. 60 (2), pp. 147-220, **TD No. 377 (October 2000)**.
- C. MICHELACCI and P. ZAFFARONI, (*Fractional*) Beta convergence, Journal of Monetary Economics, Vol. 45, pp. 129-153, **TD No. 383 (October 2000)**.

R. DE BONIS and A. FERRANDO, *The Italian banking structure in the nineties: testing the multimarket contact hypothesis*, Economic Notes, Vol. 29 (2), pp. 215-241, **TD No. 387 (October 2000)**.

```
2001
```

- M. CARUSO, Stock prices and money velocity: A multi-country analysis, Empirical Economics, Vol. 26 (4), pp. 651-72, TD No. 264 (February 1996).
- P. CIPOLLONE and D. J. MARCHETTI, *Bottlenecks and limits to growth: A multisectoral analysis of Italian industry*, Journal of Policy Modeling, Vol. 23 (6), pp. 601-620, **TD No. 314 (August 1997)**.
- P. CASELLI, *Fiscal consolidations under fixed exchange rates*, European Economic Review, Vol. 45 (3), pp. 425-450, **TD No. 336 (October 1998)**.
- F. ALTISSIMO and G. L. VIOLANTE, Nonlinear VAR: Some theory and an application to US GNP and unemployment, Journal of Applied Econometrics, Vol. 16 (4), pp. 461-486, TD No. 338 (October 1998).
- F. NUCCI and A. F. POZZOLO, *Investment and the exchange rate*, European Economic Review, Vol. 45 (2), pp. 259-283, **TD No. 344 (December 1998)**.
- L. GAMBACORTA, On the institutional design of the European monetary union: Conservatism, stability pact and economic shocks, Economic Notes, Vol. 30 (1), pp. 109-143, **TD No. 356 (June 1999)**.
- P. FINALDI RUSSO and P. ROSSI, Credit costraints in italian industrial districts, Applied Economics, Vol. 33 (11), pp. 1469-1477, TD No. 360 (December 1999).
- A. CUKIERMAN and F. LIPPI, *Labor markets and monetary union: A strategic analysis*, Economic Journal, Vol. 111 (473), pp. 541-565, **TD No. 365 (February 2000)**.
- G. PARIGI and S. SIVIERO, An investment-function-based measure of capacity utilisation, potential output and utilised capacity in the Bank of Italy's quarterly model, Economic Modelling, Vol. 18 (4), pp. 525-550, TD No. 367 (February 2000).
- F. BALASSONE and D. MONACELLI, *Emu fiscal rules: Is there a gap?*, in: M. Bordignon and D. Da Empoli (eds.), Politica fiscale, flessibilità dei mercati e crescita, Milano, Franco Angeli, **TD No. 375 (July 2000)**.
- A. B. ATKINSON and A. BRANDOLINI, Promise and pitfalls in the use of "secondary" data-sets: Income inequality in OECD countries, Journal of Economic Literature, Vol. 39 (3), pp. 771-799, TD No. 379 (October 2000).
- D. FOCARELLI and A. F. POZZOLO, The determinants of cross-border bank shareholdings: An analysis with bank-level data from OECD countries, Journal of Banking and Finance, Vol. 25 (12), pp. 2305-2337, TD No. 381 (October 2000).
- M. SBRACIA and A. ZAGHINI, *Expectations and information in second generation currency crises models*, Economic Modelling, Vol. 18 (2), pp. 203-222, **TD No. 391 (December 2000)**.
- F. FORNARI and A. MELE, Recovering the probability density function of asset prices using GARCH as diffusion approximations, Journal of Empirical Finance, Vol. 8 (1), pp. 83-110, TD No. 396 (February 2001).
- P. CIPOLLONE, *La convergenza dei salari manifatturieri in Europa*, Politica economica, Vol. 17 (1), pp. 97-125, **TD No. 398 (February 2001)**.
- E. BONACCORSI DI PATTI and G. GOBBI, The changing structure of local credit markets: Are small businesses special?, Journal of Banking and Finance, Vol. 25 (12), pp. 2209-2237, TD No. 404 (June 2001).
- G. MESSINA, Decentramento fiscale e perequazione regionale. Efficienza e redistribuzione nel nuovo sistema di finanziamento delle regioni a statuto ordinario, Studi economici, Vol. 56 (73), pp. 131-148, TD No. 416 (August 2001).

- R. CESARI and F. PANETTA, Style, fees and performance of Italian equity funds, Journal of Banking and Finance, Vol. 26 (1), TD No. 325 (January 1998).
- L. GAMBACORTA, Asymmetric bank lending channels and ECB monetary policy, Economic Modelling, Vol. 20 (1), pp. 25-46, **TD No. 340 (October 1998)**.
- C. GIANNINI, "Enemy of none but a common friend of all"? An international perspective on the lender-oflast-resort function, Essay in International Finance, Vol. 214, Princeton, N. J., Princeton University Press, TD No. 341 (December 1998).
- A. ZAGHINI, Fiscal adjustments and economic performing: A comparative study, Applied Economics, Vol. 33 (5), pp. 613-624, TD No. 355 (June 1999).
- F. ALTISSIMO, S. SIVIERO and D. TERLIZZESE, *How deep are the deep parameters?*, Annales d'Economie et de Statistique, (67/68), pp. 207-226, **TD No. 354 (June 1999)**.
- F. FORNARI, C. MONTICELLI, M. PERICOLI and M. TIVEGNA, The impact of news on the exchange rate of the lira and long-term interest rates, Economic Modelling, Vol. 19 (4), pp. 611-639, TD No. 358 (October 1999).
- D. FOCARELLI, F. PANETTA and C. SALLEO, *Why do banks merge?*, Journal of Money, Credit and Banking, Vol. 34 (4), pp. 1047-1066, **TD No. 361 (December 1999)**.
- D. J. MARCHETTI, Markup and the business cycle: Evidence from Italian manufacturing branches, Open Economies Review, Vol. 13 (1), pp. 87-103, TD No. 362 (December 1999).
- F. BUSETTI, *Testing for stochastic trends in series with structural breaks*, Journal of Forecasting, Vol. 21 (2), pp. 81-105, TD No. 385 (October 2000).
- F. LIPPI, *Revisiting the Case for a Populist Central Banker*, European Economic Review, Vol. 46 (3), pp. 601-612, **TD No. 386 (October 2000)**.
- F. PANETTA, *The stability of the relation between the stock market and macroeconomic forces*, Economic Notes, Vol. 31 (3), **TD No. 393 (February 2001)**.
- G. GRANDE and L. VENTURA, Labor income and risky assets under market incompleteness: Evidence from Italian data, Journal of Banking and Finance, Vol. 26 (2-3), pp. 597-620, TD No. 399 (March 2001).
- A. BRANDOLINI, P. CIPOLLONE and P. SESTITO, *Earnings dispersion, low pay and household poverty in Italy, 1977-1998*, in D. Cohen, T. Piketty and G. Saint-Paul (eds.), The Economics of Rising Inequalities, pp. 225-264, Oxford, Oxford University Press, **TD No. 427 (November 2001)**.
- L. CANNARI and G. D'ALESSIO, *La distribuzione del reddito e della ricchezza nelle regioni italiane*, Rivista Economica del Mezzogiorno (Trimestrale della SVIMEZ), Vol. XVI(4), pp. 809-847, Il Mulino, **TD No. 482 (June 2003)**.

2003

- F. SCHIVARDI, *Reallocation and learning over the business cycle*, European Economic Review, , Vol. 47 (1), pp. 95-111, **TD No. 345 (December 1998)**.
- P. CASELLI, P. PAGANO and F. SCHIVARDI, Uncertainty and slowdown of capital accumulation in Europe, Applied Economics, Vol. 35 (1), pp. 79-89, **TD No. 372 (March 2000).**
- P. PAGANO and G. FERRAGUTO, Endogenous growth with intertemporally dependent preferences, Contribution to Macroeconomics, Vol. 3 (1), pp. 1-38, **TD No. 382 (October 2000).**
- P. PAGANO and F. SCHIVARDI, *Firm size distribution and growth*, Scandinavian Journal of Economics, Vol. 105(2), pp. 255-274, **TD No. 394 (February 2001)**.
- M. PERICOLI and M. SBRACIA, *A Primer on Financial Contagion*, Journal of Economic Surveys, **TD No.** 407 (June 2001).
- M. SBRACIA and A. ZAGHINI, *The role of the banking system in the international transmission of shocks*, World Economy, **TD No. 409 (June 2001)**.

- E. GAIOTTI and A. GENERALE, Does monetary policy have asymmetric effects? A look at the investment decisions of Italian firms, Giornale degli Economisti e Annali di Economia, Vol. 61 (1), pp. 29-59, TD No. 429 (December 2001).
- L. GAMBACORTA, *The Italian banking system and monetary policy transmission: evidence from bank level data*, in: I. Angeloni, A. Kashyap and B. Mojon (eds.), Monetary Policy Transmission in the Euro Area, Cambridge, Cambridge University Press, **TD No. 430 (December 2001).**
- M. EHRMANN, L. GAMBACORTA, J. MARTÍNEZ PAGÉS, P. SEVESTRE and A. WORMS, *Financial systems and the role of banks in monetary policy transmission in the euro area*, in: I. Angeloni, A. Kashyap and B. Mojon (eds.), Monetary Policy Transmission in the Euro Area, Cambridge, Cambridge University Press, **TD No. 432 (December 2001)**.
- F. SPADAFORA, Financial crises, moral hazard and the speciality of the international market: further evidence from the pricing of syndicated bank loans to emerging markets, Emerging Markets Review, Vol. 4 (2), pp. 167-198, TD No. 438 (March 2002).
- D. FOCARELLI and F. PANETTA, Are mergers beneficial to consumers? Evidence from the market for bank deposits, American Economic Review, Vol. 93 (4), pp. 1152-1172, **TD No. 448 (July 2002)**.
- E.VIVIANO, Un'analisi critica delle definizioni di disoccupazione e partecipazione in Italia, Politica Economica, Vol. 19 (1), pp. 161-190, TD No. 450 (July 2002).
- F. BUSETTI and A. M. ROBERT TAYLOR, Testing against stochastic trend and seasonality in the presence of unattended breaks and unit roots, Journal of Econometrics, Vol. 117 (1), pp. 21-53, TD No. 470 (February 2003).

2004

- P. CHIADES and L. GAMBACORTA, *The Bernanke and Blinder model in an open economy: The Italian case,* German Economic Review, Vol. 5 (1), pp. 1-34, **TD No. 388 (December 2000)**.
- M. PAIELLA, *Heterogeneity in financial market participation: appraising its implications for the C-CAPM*, Review of Finance, Vol. 8, pp. 1-36, **TD No. 473 (June 2003)**.
- E. BONACCORSI DI PATTI and G. DELL'ARICCIA, *Bank competition and firm creation*, Journal of Money Credit and Banking, Vol. 36 (2), pp. 225-251, **TD No. 481 (June 2003)**.
- R. GOLINELLI and G. PARIGI, *Consumer sentiment and economic activity: a cross country comparison*, Journal of Business Cycle Measurement and Analysis, Vol. 1 (2), **TD No. 484 (September 2003)**.

FORTHCOMING

- A. F. POZZOLO, Research and development regional spillovers, and the localisation of economic activities, The Manchester School, **TD No. 331 (March 1998)**.
- F. LIPPI, Strategic monetary policy with non-atomistic wage-setters, Review of Economic Studies, TD No. 374 (June 2000).
- P. ANGELINI and N. CETORELLI, *Bank competition and regulatory reform: The case of the Italian banking industry*, Journal of Money, Credit and Banking, **TD No. 380 (October 2000)**.
- L. DEDOLA AND F. LIPPI, *The Monetary Transmission Mechanism: Evidence from the industry Data of Five* OECD Countries, European Economic Review, **TD No. 389 (December 2000)**.
- M. BUGAMELLI and P. PAGANO, *Barriers to Investment in ICT*, Applied Economics, **TD No. 420 (October 2001)**.
- D. J. MARCHETTI and F. NUCCI, Price Stickiness and the Contractionary Effects of Technology Shocks, European Economic Review, TD No. 392 (February 2001).
- G. CORSETTI, M. PERICOLI and M. SBRACIA, *Correlation analysis of financial contagion: what one should know before running a test*, Journal of International Money and Finance, **TD No. 408 (June 2001)**.
- D. FOCARELLI, Bootstrap bias-correction procedure in estimating long-run relationships from dynamic

panels, with an application to money demand in the euro area, Economic Modelling, **TD No. 440** (March 2002).

- A. BAFFIGI, R. GOLINELLI and G. PARIGI, *Bridge models to forecast the euro area GDP*, International Journal of Forecasting, **TD No. 456 (December 2002)**.
- F. CINGANO and F. SCHIVARDI, *Identifying the sources of local productivity growth*, Journal of the European Economic Association, **TD No. 474 (June 2003)**.
- E. BARUCCI, C. IMPENNA and R. RENÒ, *Monetary integration, markets and regulation*, Research in Banking and Finance, **TD No. 475 (June 2003)**.
- G. ARDIZZI, Cost efficiency in the retail payment networks: first evidence from the Italian credit card system, Rivista di Politica Economica, **TD NO. 480 (June 2003)**.
- L. GAMBACORTA and P. E. Mistrulli, *Does bank capital affect lending behavior?*, Journal of Financial Intermediation, **TD No. 486 (September 2003)**.